UNIVERSITY^{OF} BIRMINGHAM University of Birmingham Research at Birmingham

Hepatic-targeted RNA interference provides robust and persistent knockdown of alpha-1 antitrypsin levels in ZZ patients

Turner, Alice; Stolk, Jan; Bals, Robert; Linkliter, Jason; Hamilton, James; Christianson, Dawn; Given, Bruce; Burdon, Jonathan; Loomba, Rohit; Stoller, James; Teckman, Jeffery

DOI: 10.1016/j.jhep.2018.03.012

License: Creative Commons: Attribution-NonCommercial-NoDerivs (CC BY-NC-ND)

Document Version Peer reviewed version

Citation for published version (Harvard):

Turner, A, Štolk, J, Bals, R, Linkliter, J, Hamilton, J, Christianson, D, Given, B, Burdon, J, Loomba, R, Stoller, J & Teckman, J 2018, 'Hepatic-targeted RNA interference provides robust and persistent knockdown of alpha-1 antitrypsin levels in ZZ patients', *Journal of Hepatology*, vol. 69, no. 2, pp. 378-384. https://doi.org/10.1016/j.jhep.2018.03.012

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?) •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.

Accepted Manuscript

Hepatic-targeted RNA interference provides persistent knockdown of alpha-1 antitrypsin levels in ZZ patients

Alice M. Turner, Jan Stolk, Robert Bals, Jason Lickliter, James Hamilton, Dawn R. Christianson, Bruce D. Given, Jonathan G. Burdon, Rohit Loomba, James K. Stoller, Jeffery H. Teckman

PII: DOI: Reference:	S0168-8278(18)30176-4 https://doi.org/10.1016/j.jhep.2018.03.012 JHEPAT 6907
To appear in:	Journal of Hepatology
Received Date: Revised Date: Accepted Date:	31 October 201719 February 20188 March 2018



Please cite this article as: Turner, A.M., Stolk, J., Bals, R., Lickliter, J., Hamilton, J., Christianson, D.R., Given, B.D., Burdon, J.G., Loomba, R., Stoller, J.K., Teckman, J.H., Hepatic-targeted RNA interference provides persistent knockdown of alpha-1 antitrypsin levels in ZZ patients, *Journal of Hepatology* (2018), doi: https://doi.org/10.1016/j.jhep.2018.03.012

This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

Hepatic-targeted RNA interference provides persistent knockdown of alpha-1 antitrypsin levels in ZZ patients

Alice M. Turner¹, Jan Stolk², Robert Bals³, Jason Lickliter⁴, James Hamilton⁵, Dawn R. Christianson⁶, Bruce D. Given⁷, Jonathan G. Burdon⁸, Rohit Loomba⁹, James K. Stoller¹⁰, Jeffery H. Teckman¹¹

¹University of Birmingham, Birmingham, UK, A.M.Turner@bham.ac.uk
²Leiden University Medical Center, Leiden, The Netherlands, J.Stolk@lumc.nl
³Saarland University Hospital, Homburg, Germany, Robert.Bals@uniklinikum-saarland.de
⁴Nucleus Network, Melbourne, Victoria, Australia, J.Lickliter@nucleusnetwork.com.au
⁵Arrowhead Pharmaceuticals, Pasadena, CA, USA, jhamilton@arrowheadpharma.com
⁶Arrowhead Pharmaceuticals, Pasadena, CA, USA, dchristianson@arrowheadpharma.com
⁷Arrowhead Pharmaceuticals, Pasadena, CA, USA, bgiven@arrowheadpharma.com
⁸Saint Vincent's Private Hospital, Melbourne, Australia, jburdon@bigpond.net.au
⁹University of California at San Diego, San Diego, CA. USA, roloomba@ucsd.edu
¹⁰Cleveland Clinic, Cleveland, OH. USA, stollej@ccf.org
¹¹St. Louis University School of Medicine, St. Louis, MO. USA, Jeff.Teckman@health.slu.edu

Clinical Trial Registration Number: NCT02363946

Corresponding author:

Alice M. Turner

Senior Lecturer

Institute for Applied Health Research

University of Birmingham

Birmingham, UK, B15 2TT

A.M.Turner@bham.ac.uk

+44 (0) 121 3713885

+44 (0)7825683519

Keywords: siRNA, phase 1, PiZZ, RNAi, therapeutic

Word Count: 5619

1 Figure, 3 Tables

Competing interests:

J.H., B.D.G. and D.R.C. are employees of Arrowhead Pharmaceuticals. A.M.T has received grant funding from Arrowhead Pharmaceuticals and the Alpha-1 foundation. R.B. has received grant funding from Schwiete Foundation and is on advisory committees for AstraZeneca, Glaxo Smithkline and Boehringer Ingelheim. J.G.B has received compensation as a consultant from Commonwealth Serum Laboratories and Arrowhead Pharmaceuticals. J.K.S has received grant funding from CSL Behring and is a consultant for Arrowhead Pharmaceuticals, CSL Behring, Shire Pharmaceuticals, Vertex, 23andMe and is on advisory committees for Grifols and the Alpha-1 Foundation. J.H.T has received grant funding from

Alnylam, Arrowhead Pharmaceuticals, Alpha-1 Foundation, Gilead Pharmaceuticals and has been a consultant for Dicerna, Ionis Pharmaceuticals, GenKyoTex, The Alpha-1 Project, RxCelerate, Editas, Intelia, AstraZeneca, Vertex and Proteostasis. R.L. is on advisory committees or review panels for: Galmed Inc, Tobira Inc, Arrowhead Pharmaceuticals Inc; has provided consulting for: Gilead Inc, Corgenix Inc, Janssen and Janssen Inc, Zafgen Inc, Celgene Inc, Alnylam Inc, Inanta Inc, Deutrx Inc; has received grant/research support: Daiichi Sankyo Inc, AGA, Merck Inc, Promedior Inc, Kinemed Inc, Immuron Inc, Adheron Inc.

This study was funded by Arrowhead Pharmaceuticals Inc.

Author contributions: A.M.T., J.H. and D.R.C drafted the manuscript.; A.M.T, J.S., R.B., J.L. performed the clinical trial, analyzed clinical data and reviewed the manuscript; J.G.B., R.L., J.K.S. and J.H.T. oversaw the clinical trial and reviewed the manuscript; B.D.G., J.H. and D.R.C. initially designed the clinical trial protocol, analyzed clinical data and reviewed the manuscript. A.M.T., B.D.G., J.H. and D.R.C. had access to all of the data and can vouch for the integrity of the data analyses.

Abstract

Background & Aims: Alpha-1 antitrypsin deficiency (AATD) is a genetic disorder causing pulmonary and liver disease. The PiZ mutation results in mis-folded alpha-1 antitrypsin protein (Z-AAT) leading to hepatocyte accumulation, fibrosis and cirrhosis. RNAi-based therapeutics silencing production of hepatic Z-AAT might benefit patients with AATD-associated liver disease. This study evaluated an RNAi therapeutic to silence production of alpha-1 antitrypsin.

Methods: Part A of this double-blind first-in-human study randomized 54 healthy volunteers (HVs) into single dose cohorts (2 placebo: 4 active), receiving escalating doses of the investigational agent ARC-AAT from 0.38 to 8.0 mg/kg or placebo. Part B randomized 11 PiZZ genotype AATD patients who received up to 4.0 mg/kg or placebo. Patients with baseline FibroScan® >11 kPa or FEV₁ (forced expiratory volume in one second) < 60% were excluded. Assessments included safety, pharmacokinetics, and change in serum alpha-1 antitrypsin (AAT) concentrations.

Results: 36 healthy volunteers received ARC-AAT and 18 received placebo (Part A). Seven PiZZ individuals received ARC-AAT and 4 received placebo (Part B). A dose-response in serum AAT reduction was observed at doses \geq 4 mg/kg with similar relative reductions in PiZZ patients and HVs at 4 mg/kg and a maximum reduction of 76.1% (HVs) vs. 78.8% (PiZZ) at this dose. Time for serum AAT return to baseline was similar for HV and PiZZ. There were no notable differences between HV and PiZZ safety parameters. The study was terminated early due to toxicity findings related to the delivery vehicle (ARC-EX1) seen in a non-human primate study.

Conclusion: PiZZ and HV responded similarly to ARC-AAT. Deep and durable knockdown of hepatic AAT production based on observed reduction in serum AAT concentrations was demonstrated.

Lay Summary: Accumulation of abnormal proteins in the livers of patients with alpha-1 antitrypsin deficiency may lead to decreased liver function and potentially liver failure. Therapeutics targeting the production of these abnormal proteins may be used to prevent or treat liver disease in patients with alpha-1 antitrypsin deficiency.

List of Abbreviations: AATD, alpha-1 antitrypsin deficiency; Z-AAT, mutant mis-folded protein expressed by Z-allele; HVs, healthy volunteer; ARC-AAT, RNA therapeutic targeting alpha-1 antitrypsin; FEV1, forced expiratory volume in one second; AAT, alpha-1 antitrypsin; ALT, alanine aminotransferase; AST, aspartate aminotransferase; ARC-EX1, excipient used in ARC-AAT to facilitate siRNA delivery; RNAi, ribonucleic acid interference; BMI, body mass index; ER, endoplasmic reticulum; KD, knockdown; siRNA, short interfering ribonucleic acid; PK, pharmacokinetics; PBO, placebo; DLT, dose limiting toxicity; AE, adverse event; ECG, electrocardiogram; VC, vital capacity; DLCO, diffusing capacity of the lungs for carbon monoxide; ATS, American Thoracic Society; ERS, European Respiratory Society; SAE, serious adverse event; API, active pharmaceutical ingredient; LC/MS/MS, liquid chromatography-mass spectrometry; API-AAT, RNA component of ARC-AAT; chol-UNA, cholesterol-conjugated unlocked nucleobase analog; mRNA, messenger RNA; SOC, system organ class; MLP, mellitin like peptide; Cmax, maximum serum concentraction; Tmax, time to reach Cmax; AUC0-24, area under the curve from 0-24 hours; AUCinf, area under the curve time 0 to infinity; t1/2, half-life; PD, pharmacodynamic; ELF, enhanced liver fibrosis panel

Introduction

Alpha-1 antitrypsin deficiency (AATD) is an autosomal co-dominant disease which predisposes to early-onset emphysema and liver cirrhosis [1]. Estimates of the number of adult AATD patients exhibiting clinically significant liver disease have varied from 10% (defined by transaminase abnormalities) [2, 3] to 18% in a U.K. cohort assessed by ultrasound (+/- biopsy) [4] to approximately 35% in those with an Ishak fibrosis score ≥ 2 assessed histologically after liver biopsy [5]. A Swedish autopsy series from AATD patients with the PiZZ genotype found liver disease to be present in the majority of patients, albeit unrecognized in many cases during life [6]. In common with other causes of cirrhosis, many factors are thought to affect clinical presentation, including alcohol use [7], BMI [8], gender [9] and genetic modifiers. Liver disease shows a bimodal clinical presentation. Prolonged neonatal jaundice is common in newborns with the PiZZ genotype, with most recovering but others progressing to cirrhotic disease that requires transplantation at a young age [10, 11]. A second peak occurs in adults who can present with clinical liver disease that can also progress to cirrhosis and transplant, even in the absence of a childhood history of liver disease [3]. No definitive methods to predict progression to cirrhosis exist at present in either the pediatric or adult populations. Current management of liver disease in AATD is largely supportive, with transplantation the only definitive treatment. In the U.S., AATD accounts for approximately 1.5% of liver transplants with ~76 transplants performed annually [12] with outcomes similar to other common indications for transplant [13].

The most common genetic variant of alpha-1 antitrypsin (AAT) associated with liver disease is the PiZ allele whose protein product misfolds and forms polymers that are retained within the endoplasmic reticulum (ER) of hepatocytes and can be seen as periodic acid-Schiff-positive inclusions in biopsies of AATD livers [14]. Drugs targeting polymerization,

as well as those that influence cellular homeostasis mechanisms which target disposal of the mutant protein aggregates (such as autophagy) represent potential novel therapeutic routes [14]. Another route to reduce AAT polymers would be to limit production of alpha-1 antitrypsin protein by interfering with its mRNA using antisense or RNA interference target knockdown (KD) approaches. This KD method has demonstrated reduced accumulation of mutant protein globules and inhibited expression of fibrosis-associated genes in the PiZZ transgenic mouse [15]. We report here results of a randomized, placebo-controlled phase 1 study in healthy volunteers (HVs) and PiZZ AATD individuals administered ARC-AAT, a short interfering RNA (siRNA) therapeutic targeting liver production of AAT. Preclinical animal studies with ARC-AAT will be reported elsewhere.

Materials and Methods

This was a multi-center, randomized, placebo-controlled, double-blind, single doseescalation, first-in-human, Phase 1 study (NCT02363946) conducted to evaluate the safety, tolerability, pharmacokinetics (PK) and effect on circulating AAT levels of the investigational product, ARC-AAT, administered intravenously to healthy volunteers (Part A) and to PiZZ patients with AATD (Part B). Part A was initiated in HVs and Part B in individuals with AATD. With the intention of not enrolling rare disease patients at doses unlikely to have activity, Part B proceeded once target serum AAT KD of at least 30% in three of six HVs in a Part A cohort was seen, together with satisfactory safety data. The design is summarized in Fig. S1, S2. The study was approved by the appropriate institutional review committee at each site prior to commencement. Written informed consent was obtained from each subject included in the study and the study protocol conforms to the ethical guidelines of the 1975 Declaration of Helsinki as reflected in *a priori* approval by

each institution's human research committee. Participants who met all eligibility criteria (see Supplementary methods) were randomized using block randomization at a ratio of 2:1 (active:placebo) to receive a single intravenous injection of either placebo (PBO) or ARC-AAT in double-blind fashion. The study allocation sequence was generated and patients were assigned to study groups by an un-blinded statistician at the contract research organization (CPR Pharma, Adelaide, Australia). The study investigator at each site enrolled all study participants. Once screened, eligible participants were allocated to a sequentially numbered treatment and assigned a unique randomization number in accordance with the randomization schedule. The study protocol specified that all study participants and investigators remained blinded to treatment group throughout the study. The sponsor was permitted to unblind on a cohort by cohort basis after all subjects in a cohort completed their final study visit. The first study participant was enrolled December 5th, 2014 and the last participant follow up visit was November 18th, 2016.

Up to thirteen cohorts (6 participants per cohort) were planned, including 9 HV cohorts, with dosing starting at 0.38 mg/kg and escalating through doses of 1.0, 2.0, 3.0, 4.0, 5.0, 6.0, 7.0 and 8.0 mg/kg. For each participant (both Part A and Part B), clinical facility confinement was approximately 2 days, starting on Day 1 with dosing on Day 1 and discharge on Day 2. Two sentinel HV participants were dosed at 0.38 mg/kg (one ARC-AAT, one PBO) in part A; when no significant safety concerns were raised following Day 3 evaluation in these participants, the remaining four participants in Cohort 1 were treated. Part A participant data were collected at a single Phase 1 facility (Nucleus Networks, Melbourne Australia).

Part B of the study was initiated at the 2 mg/kg dose level with planned PiZZ patient cohorts at 2.0, 4.0, 6.0 and 8.0 mg/kg. Escalation to the next dose level proceeded based on all available safety data through Day 8 of the current cohort. Provision was made to halt dose

escalation if a dose-limiting toxicity (DLT) occurred, defined as a treatment-related adverse event (AE; rated as possibly or probably related) that may have posed an unacceptable risk to other participants or if target KD parameters were achieved. Part B participant data were collected during patient visits to participating clinical trial sites. Dose escalation through all HV cohorts was completed. In PiZZ individuals, the 2.0 mg/kg cohort was completed and 5 of 6 PiZZ individuals were enrolled into the 4 mg/kg cohort prior to termination of the study.

Study assessments

The safety analysis included all patients who received ARC-AAT or placebo. Safety measures included: (1) AEs; (2) physical examination; (3) vital signs; (4) 12-lead ECG measurement; (5) ECG telemetry (8 hours pre-dose to 24 hours post-dose); (6) pulmonary function testing (spirometry to include VC, FEV₁, FEV₁/VC) and DLCO, both conducted in accordance with ATS-ERS guidelines; (7) clinical laboratory tests (hematology, biochemistry, coagulation, urinalysis); (8) use of concomitant medications, and (9) recording reasons for treatment discontinuation. Scheduled evalutaions were performed by a blinded member of the site staff. Abnormalities in laboratory findings or other assessments that were deemed clinically significant by the study investigator and were initially detected during the study or were present at baseline and significantly worsened during the study were reported as AEs, whether or not they were considered drug-related by the study investigator. All AEs and SAEs were followed until resolution, until the condition stabilized, until the event was otherwise explained, or until the subject was lost to follow-up. Quantitative serum AAT was assessed by turbidometric and/or nephelometric methods (Roche Cobas[®] c501, ARCHITECT[®] ci16200TM, BNTM II System). Serum AAT reduction is based on each

individual's change from their baseline value with baseline calculated as the geometric mean

9

of three serum AAT measurements prior to receiving test article. Further details on when each assay was used and a comparison between platforms can be found in Figs. S3-S5. PK samples were analyzed using a validated hybridization-ligation method for the API and a validated LC/MS/MS method for the delivery excipient.

Investigational medicinal product

ARC-AAT Injection comprised API-AAT, the Active Pharmaceutical Ingredient consisting of a single cholesterol-conjugated unlocked nucleobase analog (chol-UNA) RNAi trigger targeting AAT mRNA supplied in a liquid solution, plus the delivery excipient ARC-EX1 containing hepatocyte-targeted N-acetylgalactosamine-conjugated melittin-like peptide supplied as a powder. The vials were prepared by an un-blinded pharmacist in the pharmacy under sterile conditions to yield a 2:1 ratio by weight (API-AAT:ARC-EX1) and then administered by a blinded site staff member as a single intravenous infusion at a rate of 20 mg/min. All subjects were treated with an oral antihistamine approximately 2 hours prior to infusion.

Objectives

Primary objectives were to determine the safety and tolerability of escalating doses of ARC-AAT, to evaluate PK of ARC-AAT components at different doses, and to determine the effect of ARC-AAT on circulating levels of AAT. Key secondary objectives included determining the dose levels achieving >30% KD and \geq 90% KD of AAT on or before Day 22 (± 1 day) when compared to the pre-dose geometric mean baseline AAT level in HVs, determining the dose level achieving > 90% KD at Day 29 (± 1 day) in 50% of PiZZ

individuals and the time for serum AAT levels to return to baseline ($\pm 15\%$ from geometric mean baseline or > 90 mg/dL for HV).

Data handling and statistical analysis

No statistical methods were used to determine cohort sizes or total subjects planned for enrollment. Safety results were summarized by cohort and treatment group. The frequency of treatment-emergent AEs, SAEs, related AEs, related SAEs, and AEs leading to withdrawal, dose modification, or treatment discontinuation were summarized by dose and treatment group according to the latest version of MedDRA by System Organ Class (SOC) and Preferred Term. Plasma concentrations of ARC-AAT Injection constituents (MLP, chol-UNA) following a single dose of ARC-AAT injection at different dose levels were used to calculate the following ARC-AAT PK parameters: Cmax, tmax, AUC0-24, AUCinf, and t1/2. Pharmacokinetic parameters were determined using non-compartmental method(s) with Phoenix[™] WinNonlin® Version 6.3 software (Copyright ©1998-2012, Certara L.P., distributed by Pharsight Corporation). Descriptive statistics of PK parameters included mean, standard deviation, and coefficient of variation. Pharmacodynamic (PD) analyses were performed using Microsoft Excel 2013 and GraphPad Prism software version 6.0. Descriptive statistics included geometric mean, mean, standard error of mean, and percent AAT serum concentration change from baseline.

Results

Healthy volunteer and PiZZ individual characteristics

Fifty-four HVs were enrolled with 36 receiving ARC-AAT and 18 receiving placebo. All HVs resided in Australia; characteristics are shown in Table 1. Eleven PiZZ AATD individuals were enrolled from patients screened in the United Kingdom, Australia, the Netherlands and Germany; their characteristics are shown in Table 1. PiZZ patients included in the study had minimal or no liver or lung disease. Baseline FEV1 and FibroScan® parameters for individual PiZZ patients are presented in Table 2. The mean PiZZ patient FEV1 at baseline was 85% (Range 72-93%). The mean baseline FibroScan® was 5.58 kPa (Range 4.0-6.6) which is consistent with very minimal or no liver fibrosis. Seven PiZZ individuals received ARC-AAT and 4 received placebo. Fig. S2 presents the Consort diagram for the study.

Safety, pharmacokinetics and pharmacodynamics

In both Part A (HV) and Part B (PiZZ), there were no deaths and no serious adverse events (SAEs) in subjects receiving ARC-AAT. There was one SAE of rhabdomyolysis in a subject receiving placebo. There were no AEs rated as severe in intensity in subjects receiving ARC-AAT or AEs leading to premature study discontinuation. The only AE showing a pattern indicating possible dose relatedness was chills, reported in 3 of 4 HVs at the 8 mg/kg dose and likely related to the infusion of ARC-AAT (Tables S1A, B). Headaches, muscle spasms, upper respiratory tract infection, nausea and dysmenorrhea were seen more frequently in healthy volunteer subjects receiving ARC-AAT. However, there is no clear mechanistic association between ARC-AAT and these symptoms. In the AATD patients, no AE occurred more than once in the seven subjects receiving ARC-AAT. A single AE with the preferred term of "asthma" was reported in an AATD patient receiving 2 mg/kg of ARC-AAT. This AE was reported as mild and in a patient with a previous history of asthma treated with

inhaled steroids and inhaled β2-receptor agonists. Symptoms began on Day 7 and were reported to last approximately 5 minutes, then resolving with use of inhaled beta-agonists. The study investigator reported this AE as not related to ARC-AAT but rather related to cold air exposure. This patient did not experience any meaningful adverse changes in FEV1 during the study. There were no clinically significant adverse changes in ECG, DLCO, or FEV₁ and no increased rate of pulmonary AEs in subjects receiving ARC-AAT. Baseline and post-dose values for FEV1 and select clinical laboratory assessments in HV subjects and PiZZ patients are presented in Table S2A, B. The study was terminated early due to toxicity findings related to the delivery vehicle (ARC-EX1) seen in a non-human primate study.

All subjects randomized to ARC-AAT in Part A had PK concentrations reported for all samples collected. One HV dosed at 8.0 mg/kg did not receive a full dose (<50%) and thus was excluded from the PK summary. The PK parameters in HVs and patients are shown in the Table S3; overall results indicated that the API circulates with a short half-life of approximately 4 hours. The delivery excipient, ARC-EX1, demonstrated a half-life of 8-10 hours. Exposure increased linearly with dose.

There was a clear PD dose response to ARC-AAT when considering serum AAT levels, with prominent (>30%) reductions seen at all doses $\geq 2mg/kg$ (Fig. 1; Fig. S6). The greatest reductions were seen at approximately four weeks post-dose. The time to reach maximum serum AAT reduction may be in part due to the 5-7 day half-life of serum AAT. Reductions in serum AAT levels achieved in Cohort 3 (2.0 mg/kg) were sufficient to transition to Part B of the study based on the criteria pre-defined in the protocol. Continuing dose escalation in HVs showed a dose-related response for reduction in serum AAT levels, with mean relative maximum reductions of 88.3% seen at the 8.0 mg/kg dose level (Fig. 1; Table 3; Fig S6). The relative percentage reduction in AAT for PiZZ patients receiving 2.0 mg/kg and 4.0 mg/kg ARC-AAT was similar to that in the healthy subjects at the same dose

level (Fig. 1). The maximum reduction in a single PiZZ patient was 78.8% at 4.0 mg/kg ARC-AAT, compared with 76.1% in HVs at this dose. Although the percentage change in serum AAT changes were similar in HV and PiZZ subjects, the baseline and maximum post-dose reduction absolute values were different. The average baseline serum AAT concentration for HV was 144.3 mg/dL (range: 101-216) and for PiZZ was 25.3 mg/dL (range: 20.3-35.0) (Fig. S7). The lowest serum AAT concentration reached was 13.2 and 5.9 mg/dL in HV and PiZZ subjects, respectively.

Dose range to achieve target knockdown and duration of knockdown

Greater than 30% KD was achieved in HVs at doses of 3.0 mg/kg and above, while 89.8% was the maximum KD achieved at 8.0 mg/kg in any subject, with a clear dose response and plateau effect at doses \geq 6.0 mg/kg (Fig. 1; Table 3). Similar effects in PiZZ patients were observed (Fig. 1), although the 90% KD target was not achieved, presumably due to early termination of dose escalation. The duration of \geq 30% KD was around 8 weeks in HVs at optimal (\geq 6 mg/kg) doses, and KD durations were similar in PiZZ and HVs at equivalent doses (Fig. 1).

Discussion

The main finding from this randomized, controlled phase 1 study of a hepatocyte-targeted RNAi therapeutic was a clear dose response in serum AAT knockdown, with a mean maximum knockdown of 88.3% observed at the highest (8 mg/kg) dose level in healthy volunteers. A dose response was also seen in PiZZ individuals with maximum serum AAT knockdown of 78.8% at the 4 mg/kg dose level. The level of knockdown seen in healthy

volunteers and in PiZZ individuals was similar at equivalent doses. In this study, ARC-AAT was safe, with no SAEs observed in actively treated subjects while other AEs observed were relatively unremarkable (Tables S1A, B).

Circulating serum AAT levels in healthy adults will generally exceed 90 mg/dL and represent a combination of AAT secreted from the liver and extrahepatic sources [16]. Based on the plateau in maximum AAT reduction (Fig. 1), it appears that approximately 90% of circulating AAT is produced in the liver in phenotypically normal adults. This reduction is similar to that observed in transgenic mice transfected with the human PiZ AAT gene when they are treated with maximal doses of a hepatic-directed siRNA [17]. Due to the premature discontinuation of this clinical trial because of an unexpected animal toxicity seen with a different compound sharing the same ARC-EX1 delivery excipient, we were unable to evaluate the full dose range in PiZZ AATD patients as planned. As expected from the experience in PiZ transgenic mice and HVs, ARC-AAT reduced circulating AAT serum concentrations in AATD patients. Comparing percentage reductions for AATD individuals and HVs (Fig. 1; Table 3), PiZZ individuals responded at least as well to ARC-AAT with respect to both depth and duration of AAT knockdown. Similar results have been reported by other groups developing siRNA therapeutics targeting hepatic alpha-1 antitrypsin production [18].

The method used for AAT knockdown involved the active agent (API) and a delivery excipient (ARC-EX1). The API consisted of a single cholesterol-conjugated unlocked nucleobase analog (chol-UNA) targeting AAT mRNA. The RNAi trigger was conjugated to cholesterol to increase the uptake of API in the liver by endocytosis while the delivery excipient, also targeted to the liver by conjugation to N-acetylgalactosamine, enabled endosomal escape of the siRNA into the cytoplasm where RNAi occurs [19]. This approach was effective in that it resulted in significant knockdown of AAT in both HVs and PiZZ

patients. This indicates that RNAi is a viable strategy for further development for the reduction of AAT hepatic synthesis.

AATD predisposes to emphysema because of an inadequate protease screen in the lung, leaving neutrophil elastase unchecked and resulting in loss of alveolar walls. In contrast to the toxic loss of function that defines the emphysema risk in AATD, the pathogenesis of liver disease relates to a toxic gain of function related to the inability to secrete polymerized AAT that accumulates in the hepatocyte with specific alpha-1 antitrypsin variants, leading to endoplasmic reticulum stress, fibrosis, and ultimately cirrhosis. A recent cross-sectional study of adult AATD patients demonstrated that advanced fibrosis can be present without clinical manifestations of disease [5]. However, liver fibrosis and even cirrhosis may regress with removal of the hepatic insult which supports silencing hepatic alpha-1 antitrypsin production in patients with liver disease [20]. Survival in AATD patients is increasing, likely due to less smoking, reduced exposure to inhaled environmental toxins and better pulmonary care [21]. With improvement in lung disease-related mortality, it seems likely that liver disease will be increasingly recognized in PiZZ adults in the future. The only options currently available for patients with AATD-associated liver disease include standard supportive care (e.g. avoidance of hepatic toxins and vaccination) and liver transplant. The development of RNAi therapeutics targeting the hepatic production of Z-AAT protein would represent a novel approach to preventing and potentially reversing AATD-associated liver disease. The results from this first-in-human study are encouraging in that they demonstrate potent and prolonged reduction of hepatic AAT synthesis as measured by serum AAT levels. This first-in-human study of an RNAi therapeutic may represent a first step in the development of a new class of hepatocyte-targeted therapeutics for the treatment of AATD-associated liver disease.

Detecting AATD liver disease is generally based on clinical suspicion. Optimal strategies to image the liver for early detection of fibrosis in AATD, as are recognized in

16

many other fibrotic liver conditions [22], are not yet established [23]. Subjects included in our study did not have clinically significant liver disease, based on either clinical evaluation or on transient elastography (FibroScan®) at screening. This and the phase 1 single escalating dose design precludes any conclusions being drawn about the efficacy of ARC-AAT in reducing liver pathology. Future multi-dose studies of siRNA drug candidates for AATD liver disease will likely need to demonstrate a histologic benefit based on liver biopsy, a reduction in the rates of liver-related events (e.g. variceal bleed or transplant) or a major effect on surrogate outcomes such as elastography or other biomarkers. A small study of the enhanced liver fibrosis panel (ELF) as a biomarker in AATD patients has been published [24], but further work is needed to understand if this would be a viable, less invasive alternative to biopsy.

With termination of all ARC-EX1 containing programs by the study sponsor due to ARC-EX1 delivery excipient-related toxicity findings in a non-clinical study, further development of RNAi mediated knockdown of AAT is ongoing utilizing strategies which do not require a delivery excipient. In addition to an expected reduction in toxicity, this newer method in pre-clinical development can potentially use subcutaneous rather than intravenous administration.

One concern facing this strategy is the potential that knockdown of AAT may increase the risk of lung disease in AATD patients if used chronically, since participants with null genotypes who produce no AAT in any tissue exhibit more severe emphysema [25]. In this study, no significant adverse changes in lung function were seen, although we recognize that the short duration of treatment was insufficient to fully examine for changes in lung function. To mitigate concerns of RNAi-induced reductions in serum AAT in AATD participants with pre-existing emphysema, AAT augmentation therapy could be used concomitantly with liver-targeted siRNA therapeutics intended to treat liver disease [26]. It

should also be emphasized that the AATD patients treated in this study by design were relatively healthy with minimal lung or liver disease based on standard non-invasive measures. The safety and pharmacodynamic response of this approach will need to be evaluated in patients with more advanced lung and liver disease.

Strengths of our study include the large number of HVs tested, the wide range of doses studied and the inclusion of PiZZ subjects in the age group where such a product would likely be used. Limitations include the early study termination, short duration of treatment and the lack of liver biopsy outcomes in participants. However, we believe that liver biopsy would not have been appropriate prior to phase 2 work due to its invasive nature, its corresponding procedural risk, and uncertain therapeutic benefit in this single dose phase 1 study that enrolled relatively healthy PiZZ subjects.

In conclusion, RNAi can be used to reduce hepatic production of AAT based on dose responsive reductions in serum AAT levels seen after administration of single doses of ARC-AAT, and siRNA shows promise for treatment of AATD-associated liver disease.

Acknowledgments: The authors thank Vincent Wong and Gabriel Palluconi for assistance with clinical trial conduct and Sue Brenton for patient recruitment. We also thank The Alpha-1 Project for pre-clinical financial support and collaborative efforts that supported development of ARC-AAT.

REFERENCES

- [1] Stockley RA, Turner AM. Alpha-1-antitrypsin deficiency: clinical variability, assessment, and treatment. Trends Mol Med 2014;20:105-115.
- [2] Tanash HA, Nystedt-Duzakin M, Montero LC, Sveger T, Piitulainen E. The Swedish alpha1-antitrypsin screening study: health status and lung and liver function at age 34. Ann Am Thorac Soc 2015;12:807-812.
- [3] Clark V, Brantly M, Dhanasekaran R, Schreck P, Rouhani F, Nelson D. ALT abnormalities in adults with alpha-1 antitrypsin deficiency. J Hepatol 2011;54:S354.
- [4] Dawwas MF, Davies SE, Griffiths WJ, Lomas DA, Alexander GJ. Prevalence and risk factors for liver involvement in individuals with PiZZ-related lung disease. Am J Respir Crit Care Med 2013;187:502-508.
- [5] Clark VC, Marek GW, Kurtz TL, Liu C, Rouhant F, Brantly M. Liver fibrosis is present in one-third of adults with alpha-1 antitrypsin deficiency without overt liver disease. Hepatology 2015;62:279A-280A.
- [6] Elzouki AN, Eriksson S. Risk of hepatobiliary disease in adults with severe alpha 1antitrypsin deficiency (PiZZ): is chronic viral hepatitis B or C an additional risk factor for cirrhosis and hepatocellular carcinoma? Eur J Gastroenterol Hepatol 1996;8:989-994.
- [7] Triger DR, Millward-Sadler GH, Czaykowski AA, Trowell J, Wright R. Alpha-1-antitrypsin deficiency and liver in adults. Q J Med 1976;45:B51-72.
- [8] Bernspang E, Carlson J, Piitulainen E. The liver in 30-year-old individuals with alpha(1)antitrypsin deficiency. Scand J Gastroenterol 2009;44:1349-1355.
- [9] Eriksson S, Carlson J, Velez R. Risk of cirrhosis and primary liver cancer in alpha 1antitrypsin deficiency. N Engl J Med 1986;314:736-739.
- [10] Nelson DR, Teckman J, Di Bisceglie AM, Brenner DA. Diagnosis and management of patients with alpha1-antitrypsin (A1AT) deficiency. Clin Gastroenterol Hepatol 2012;10:575-580.
- [11] American Thoracic Society/European Respiratory Society statement: Standards for the diagnosis and management of individuals with alpha-1 antitrypsin deficiency. Am J Respir Crit Care Med 2003;168:818-900.
- [12] Chu AS, Chopra KB, Perlmutter DH. Is severe progressive liver disease caused by alpha-1-antitrypsin deficiency more common in children or adults? Liver Transpl 2016;22:886-894.
- [13] National Health Service Blood and Transplant. Organ donation and transplantation activity report 2014/2015. October ed. Watford, United Kingdom; 2016.
- [14] Teckman JH, Mangalat N. Alpha-1 antitrypsin and liver disease: mechanisms of injury and novel interventions. Expert Rev Gastroenterol Hepatol 2015;9:261-268.
- [15] Guo S, Booten SL, Aghajan M, Hung G, Zhao C, Blomenkamp K, et al. Antisense oligonucleotide treatment ameliorates alpha-1 antitrypsin-related liver disease in mice. J Clin Invest 2014;124:251-261.
- [16] Teckman JH. Liver disease in alpha-1 antitrypsin deficiency: current understanding and future therapy. COPD 2013;10 Suppl 1:35-43.
- [17] Wooddell CI, Peterson RM, Blomenkamp KS, Subbotin VM, Chu Q, Hamilton HL, et al. RNA interference therapeutic ARC-AAT prevents production of Z-alpha1 antitrypsin polymers and reverses liver disease phenotype in PiZ mouse model. Hepatology 2016;64:68A
- [18] Haslett P. A randomized, single-blind, placebo-controlled, phase 1/2 study of ALN-AAT, an investigational RNAi therapeutic for the treatment of alpha-1 antitrypsin deficiency associated liver disease. 12th Annual Meeting of the Oligonucleotide Therapeutics Society (OTS); Montreal, Canada: Oligonucleotide Therapeutics Society; September 28, 2016.

- [19] Wooddell CI, Rozema DB, Hossbach M, John M, Hamilton HL, Chu Q, et al. Hepatocytetargeted RNAi therapeutics for the treatment of chronic hepatitis B virus infection. Mol Ther 2013;21:973-985.
- [20] Ellis EL, Mann DA. Clinical evidence for the regression of liver fibrosis. J Hepatol 2012;56:1171-1180.
- [21] Choate R, Mannino D, Sandhaus R, Barber D, Holm K. Improving survival among ZZ patients in the Alphanet Program. Am J Respir Crit Care Med 2016;193:A1569.
- [22] Friedrich-Rust M, Ong MF, Martens S, Sarrazin C, Bojunga J, Zeuzem S, et al. Performance of transient elastography for the staging of liver fibrosis: a meta-analysis. Gastroenterology 2008;134:960-974.
- [23] Kim RG, Nguyen P, Bettencourt R, Dulai PS, Haufe W, Hooker J, et al. Magnetic resonance elastography identifies fibrosis in adults with alpha-1 antitrypsin deficiency liver disease: a prospective study. Aliment Pharmacol Ther 2016;44:287-299.
- [24] Janciauskiene S, Wallmark A, Piitulainen E, Kohnlein T, Welte T, Sveger T. Performance of enhanced liver fibrosis plasma markers in asymptomatic individuals with ZZ alpha1antitrypsin deficiency. Eur J Gastroenterol Hepatol 2011;23:716-720.
- [25] Fregonese L, Stolk J, Frants RR, Veldhuisen B. Alpha-1 antitrypsin null mutations and severity of emphysema. Respir Med 2008;102:876-884.
- [26] Edgar RG, Patel M, Bayliss S, Crossley D, Sapey E, Turner AM. Treatment of lung disease in alpha-1 antitrypsin deficiency: a systematic review. Int J Chron Obstruct Pulmon Dis 2017;12:1295-1308.

Figure Legends

Title:

Fig. 1: Serum AAT dose response to ARC-AAT in healthy volunteers and PiZZ subjects.

Legend:

A single intravenous dose of (A) 0.38 to 8.0 mg/kg ARC-AAT (Cohorts 1-3g) was given to healthy volunteers, and (B) 2 and 4 mg/kg ARC-AAT (cohorts 4 and 5) given to PiZZ subjects. Serum AAT was measured weekly for the first four-weeks and then bi-weekly until serum concentrations returned to baseline. (C) Duration of knockdown in healthy volunteers and PiZZ subjects was similar. Mean line plotted when n>1 at time point. Subjects receiving ARC-AAT per dose level, n=3-4 (healthy volunteers), n=3 (PiZZ), receiving placebo, n=18 (healthy volunteers), n=4 (PiZZ). Legend doses are in mg/kg. Data represents the mean \pm SEM. Abbreviations: PBO, Placebo; HV, healthy volunteers; ZZ, PiZZ subjects.

Table 1: Characteristics of the participants

Legend: Numerical values are shown as mean (SD) and frequency variables as n (%)

Dose (mg N = HV,		0.38 n=4	1.0 n=4	2.0 n=4, 4	3.0 n=4	4.0 n=4, 3	5.0 n=4	6.0 n=4	7.0 n=4	8.0 n=4	Total n=36, 7	Placebo n=18, 4	All n=54
HVs	Mean age in years (SD)	39.0 (9.8)	25.0 (4.5)	30.0 (7.8)	29.5 (10.7)	32.3 (5.6)	23.8 (4.1)	32.5 (15.3)	26.0 (5.0)	25.0 (1.8)	29.2 (8.6)	25.4 (4.8)	28.0 (7.7
	Mean BMI (SD)	25.28 (2.92)	21.48 (1.36)	23.10 (3.38)	23.75 (2.70)	23.45 (2.01)	21.75 (2.08)	23.85 (1.61)	22.53 (2.03)	23.65 (3.47)	23.20 (2.46)	23.38 (2.37)	23.2
	Male	3 (75%)	0	2 (50%)	3 (75%)	3 (75%)	0	1 (25%)	1 (25%)	2 (50%)	15 (42%)	6 (33%)	21 (39%
	Mean baseline serum AAT, mg/dL (SD)	(12.2)	154 (29.1)	119 (15.6)	137 (12.1)	140 (20.5)	183 (36.4)	151 (27.1)	164 (22.0)	141 (17.3)	(12/3) 145 (29.3)	(143) (23.0)	145 (27.3
PiZZ patients	Mean age in years (SD)			59.0 (1.8)		64.3 (3.8)				5	61.3 (3.8)	58.8 (9.0)	60.4 (5.9)
	Mean BMI (SD)			27.50 (3.56)		24.10 (2.63)					26.04 (3.46)	26.43 (2.73)	26.13 (3.07
	Male			1(25%)		2 (67%)						1 (25%)	4 (36
	Mean Baseline FEV1%* (Range)			85 (72- 93)		91 (78- 104)		2			87.57 (72- 104)	93.25 (80- 104)	89.64 (72-1
	Mean baseline DLCO% ** (Range)			65.75 (57-73)		63.33 (51-73)	P				64.71 (51- 73)	68.5 (43- 105)	66.09 (43-1
	Mean baseline serum AAT, mg/dL (SD)			25.2 (6.1)		27.4 (2.9)					26.2 (5.0)	24.3 (4.4)	25.5 (4.8)
	Mean baseline FibroScan® kPa (Pange)			5.58 (4.0-6.6)		5.57 (5.0- 6.1)					5.57 (4.0- 6.6)	5.58 (4.1- 7.5)	5.57 (4.0-
	FibroScan® kPa		Ś			· ·						1.3)	(4
	5												

22

Table 2: Individual PiZZ patients baseline measures of lung and liver disease severity.

Legend: FEV1% = forced	l expiratory volume in on	e second as a % of predicted
-------------------------------	---------------------------	------------------------------

Patient	Gender	Age	Baseline FEV1%	Baseline FibroScan® (kPa)	ALT (U/L) Day 1 pre-dose
Patient 1	Female	58	93	5.5	19
Patient 2	Female	64	80	5.5	17
Patient 3	Female	57	72	6.6	17
Patient 4	Male	61	84	6.2	19
Patient 5	Female	60	91	4.0	27
Patient 6	Female	59	97	5.2	23
Patient 7	Male	66	78	5.6	17
Patient 8	Female	66	92	4.1	12
Patient 9	Male	60	104	5.0	38
Patient 10	Female	67	91	6.1	17
Patient 11	Male	46	104	7.5	24
		2			

Table 3: Maximum serum AAT KD across dose levels in healthy volunteers and PiZZ

individuals

Legend: Abbreviations: KD, knockdown; HV, healthy volunteers; ZZ, PiZZ patients

Dose Level (mg/kg)	РВО	0.38	1.0	2.0	3.0	4.0	5.0	6.0	7.0	8.0
HV (n)	18	4	4	4	4	4	4	4	3	3
Max KD (%)	24.8	9.30	31.9	36.3	61.0	76.1	86.7	87.1	85.1	89.8
Mean Max KD (%) (SEM)	8.44 (1.80)	6.55 (1.49)	25.9 (3.29)	26.7 (7.14)	45.3 (6.80)	64.8 (6.07)	78.1 (4.38)	83.3 (1.93)	82.6 (1.31)	88.3 (0.769)
P value	N/A	0.6363	0.0004	0.0014	< 0.0001	< 0.0001	< 0.0001	< 0.0001	< 0.0001	< 0.0001
ZZ	4	-	-	3	-	3	-	-	-	-
Max KD (%)	46.6	-	-	73.7	-	78.8	-	-	-	-
Mean Max KD (%) (SEM)	21.9 (9.81)	-	-	44.1 (15.1)	-	78.1 (0.481)	-	-	-	-
P value	N/A	-	-	0.2524		0.0047	-	-	-	-

Article Highlights:

- Alpha-1 antitrypsin deficiency is associated with liver fibrosis and cirrhosis •
- RNAi drugs might benefit patients with alpha-1 antitrypsin deficiency •
- ARC-AAT reduced alpha-1 antitrypsin protein in study participants •

....



